\documentclass[12pt]{article}

\setlength{\parindent}{0pt}
\everymath={\displaystyle}
\pagestyle{empty}
\newcommand{\answerspace}{\vfill}
\setlength{\textwidth}{6in}
\setlength{\hoffset}{-0.5in}
\setlength{\textheight}{9in}
\setlength{\voffset}{-1in}



\begin{document}

\textbf{Math 126: Calculus II}
\hfill\makebox[2.5in][r]{Name:\hrulefill}

\textbf{Quiz 6}\quad \textsl{October 11, 2000}
\hfill\makebox[2.5in][r]{Section:\hrulefill}

\bigskip


In both problems, be sure to \emph{show all your work and include all
important steps}.

\bigskip


1. Evaluate  $\int \frac{\sqrt{1-x^2}}{x^2}\,dx$.

\answerspace

2. Evaluate $\int_0^\infty  x e^{-x^2}\,dx$.

\answerspace
 
\end{document}


